Aalborg Universitet



Antithrombotic Usage, Including Three-Year Outcomes With Dabigatran and Vitamin K Antagonists for Atrial Fibrillation, in Eastern Europe: A Descriptive Analysis From Phase 3 of the GLORIA-AF Registry

Bergler-Klein, Jutta; Gotcheva, Nina; Kalējs, Oskars; Kalarus, Zbigniew; Kovačić, Dragan; Peršić, Viktor; Shlyakhto, Evgeny; Uuetoa, Tiina; Huisman, Menno V.; Lip, Gregory Y. H.; Vinereanu, Dragos; GLORIA-AF Investigators Published in: American Journal of Therapeutics

DOI (link to publication from Publisher): 10.1097/MJT.0000000000001655

Creative Commons License CC BY-NC-ND 4.0

Publication date: 2024

Document Version Publisher's PDF, also known as Version of record

Link to publication from Aalborg University

Citation for published version (APA):

Bergler-Klein, J., Gotcheva, N., Kalējs, O., Kalarus, Z., Kovačić, D., Peršić, V., Shlyakhto, E., Uuetoa, T., Huisman, M. V., Lip, G. Y. H., Vinereanu, D., & GLORIA-AF Investigators (2024). Antithrombotic Usage, Including Three-Year Outcomes With Dabigatran and Vitamin K Antagonists for Atrial Fibrillation, in Eastern Europe: A Descriptive Analysis From Phase 3 of the GLORIA-AF Registry. *American Journal of Therapeutics*, 31(1), e1-e12. https://doi.org/10.1097/MJT.000000000001655

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- Users may download and print one copy of any publication from the public portal for the purpose of private study or research.

- You may not further distribute the material or use it for any profit-making activity or commercial gain You may freely distribute the URL identifying the publication in the public portal -

Take down policy If you believe that this document breaches copyright please contact us at vbn@aub.aau.dk providing details, and we will remove access to the work immediately and investigate your claim.

Downloaded from vbn.aau.dk on: July 02, 2025

OPEN

Antithrombotic Usage, Including Three-Year Outcomes With Dabigatran and Vitamin K Antagonists for Atrial Fibrillation, in Eastern Europe: A Descriptive Analysis From Phase 3 of the GLORIA-AF Registry

Jutta Bergler-Klein, MD,¹ Nina Gotcheva, MD, PhD,² Oskars Kalējs, MD,³ Zbigniew Kalarus, MD,⁴ Dragan Kovačić, MD,⁵ Viktor Peršić, MD, PhD,^{6,7} Evgeny Shlyakhto, MD, PhD,⁸ Tiina Uuetoa, MD,⁹ Menno V. Huisman, MD, PhD, FESC,^{10*} Gregory Y. H. Lip, MD,^{11,12*} and Dragos Vinereanu, MD, PhD^{13,14*,†} on behalf of the GLORIA-AF Investigators

Background: Global Registry on Long-Term Oral Antithrombotic Treatment in Patients with Atrial Fibrillation (GLORIA-AF) is a prospective registry of outcomes from patients with newly diagnosed AF at risk of stroke. In the propensity score (PS)-matched global population of phase 3 GLORIA-AF, at 3 years, dabigatran-treated patients experienced reduced risk for major bleeding, and similar risk for stroke and myocardial infarction, compared with vitamin K antagonist (VKA)-treated patients.

Study Question: Do patients in Eastern Europe benefit from treatment with dabigatran versus VKA?

Study Design: Descriptive analysis, without PS matching. To contextualize the Eastern Europe results of GLORIA-AF phase 3, we also descriptively analyzed the global population without PS matching. Consecutive patients with newly diagnosed AF and CHA2DS2-VASc-score \geq 1 were enrolled until December 2016 in 38 countries (9 in Eastern Europe).

Measures and Outcomes: Three-year outcomes with dabigatran and VKA.

Results: In Eastern Europe, 1341 patients were eligible (6% of patients globally), and incidence rates (per 100 patient-years) for the following outcomes were numerically lower with dabigatran (N = 498) versus VKA (N = 466): major bleeding (0.26 vs. 0.90), all-cause death (2.04 vs. 3.50), and a composite of stroke, systemic embolism, myocardial infarction, life-threatening bleeding, and vascular death (1.37 vs. 1.92); stroke was comparable (0.51 vs. 0.50). All incidence rates were numerically lower in Eastern Europe versus the global population for both treatments. Chronic concomitant use of high bleeding risk medications (eg, nonsteroidal anti-inflammatories) was lower in Eastern Europe (dabigatran 3.8%, VKA 9.3%) than globally (dabigatran 14.8%, VKA 20.6%) and persistence with dabigatran was higher in Eastern Europe (76%) than globally (64%).

Conclusions: Dabigatran was associated with numerically reduced major bleeding, all-cause death, and cardiovascular (CV) composite, with comparable risk of stroke versus VKA, in Eastern Europe. Limitations of this descriptive analysis include few CV events (n = 11 for stroke, in the dabigatran and VKA groups combined) and a lack of statistical analysis and PS matching, which precludes definitive conclusions; however, the CV outcomes in Eastern Europe were consistent with the beneficial impact of dabigatran versus VKA in the statistically analyzed global population with PS matching.

Keywords: atrial fibrillation, dabigatran, vitamin K antagonist, stroke prevention, GLORIA-AF

INTRODUCTION

Atrial fibrillation (AF) affects up to 3% of adults,¹ with lifetime risk of approximately 25% for people more than 40 years of age,² and increasing prevalence in an aging population.³ Patients with AF have a five-fold higher risk of stroke,⁴ which is a

particularly common cause of death 5 and major disability. 6

Anticoagulation with vitamin K antagonists (VKA), such as warfarin, reduces the risk of stroke and mortality.⁷ However, there are significant limitations to VKA therapy, including a narrow therapeutic margin, unpredictable dose–response, drug–drug, and drug–

¹Department of Cardiology, University Clinic of Internal Medicine II, Medical University of Vienna, Vienna, Austria; ²Department of Cardiology, National Heart Hospital, Sofia, Bulgaria; ³Department of Arrhythmology, Pauls Stradins Clinical University Hospital, Riga, Latvia; ⁴Department of Cardiology, Congenital Heart Diseases and Electrotherapy, Medical University of Silesia, Silesian Centre for Heart Diseases, Zabrze, Poland; ⁵Department of Cardiology, General Hospital Celje, Celje, Slovenia; ⁶Department of Medical Rehabilitation, Medical Faculty, University of Rijeka, Rijeka, Croatia; ⁷Division of Cardiology, Hospital for Medical Rehabilitation of the Heart and Lung Diseases and Rheumatism "Thalassotherapia Opatija," Opatija, Croatia; ⁸Clinical Endocrinology Laboratory, Department of Endocrinology, Almazov National Medical Research Centre, Saint Petersburg, Russia; ⁹Confido Healthcare Group, Tallinn, Estonia; ¹⁰Department of Thrombosis and Hemostasis, Leiden University Medical Center, Leiden, the Netherlands; ¹¹Liverpool Centre for Cardiovascular Science at University of Liverpool, Liverpool John Moores University, and Liverpool Heart & Chest Hospital, Liverpool, United Kingdom; ¹²Department of Clinical Medicine, Aalborg University, Aalborg, Denmark; ¹³University and Emergency Hospital, Bucharest, Romania.

Supported by Boehringer Ingelheim.

J. Bergler-Klein reports no conflicts of interest. N. Gotcheva reports no conflicts of interest. O. Kalējs reports grants and personal fees from Bayer and Boehringer Ingelheim. Z. Kalarus has lectured for Boehringer Ingelheim. D. Kovačić reports no conflicts of interest. V. Peršić reports no conflicts of interest. E. Shlyakhto reports no conflicts of interest. T. Uuetoa reports grants and personal fees from Bayer and Boehringer Ingelheim. M. V. Huisman reports grants from ZonMW Dutch Healthcare Fund, and grants and consultation honoraria from Boehringer Ingelheim, Pfizer/Bristol-Myers Squibb, Bayer Health Care, Aspen, and Daiichi-Sankyo. G. Y. H. Lip reports consultancy and speaking for BMS/Pfizer, Boehringer Ingelheim, and Daiichi-Sankyo. No fees are received personally. D. Vinereanu reports grants and personal fees from Bayer, Boehringer Ingelheim, and Pfizer.

Supplemental digital content is available for this article. Direct URL citations appear in the printed text and are provided in the HTML and PDF versions of this article on the journal's Web site (www.americantherapeutics.com).

All authors contributed to the writing process by reviewing drafts of the manuscript for intellectual content and interpretation. All authors approved the final draft for publication.

GLORIA-AF was conducted in accordance with the principles of Good Clinical Practice, the Declaration of Helsinki, Good Epidemiological Practice and Good Pharmacoepidemiology Practices. The protocol was approved by the European Medicines Agency and institutional review boards at each participating site. For administrative purposes the study is divided into 2 protocol numbers: 1160.129 for all non-EU (European Union) and non-EEA (European Economic Area) countries, and 1160.136 for EU and EEA countries. The protocol dates are 8 June 2011 (revised 7 June 2013) and 23 March 2012 (revised 22 October 2014), respectively.

*Drs. M. V. Huisman and G. Y. H. Lip are co-Chairs of the GLORIA-AF registry and joint senior authors with Dr. D. Vinereanu.

To ensure independent interpretation of clinical study results, Boehringer Ingelheim grants all external authors access to all relevant material, including participant-level clinical study data, and relevant material as needed by them to fulfill their role and obligations as authors under the ICMJE criteria. Furthermore, clinical study documents (eg, study report, study protocol, statistical analysis plan) and participant clinical study data are available to be shared after publication of the primary manuscript in a peer-reviewed journal and if regulatory activities are complete and other criteria met per the BI Policy on Transparency and Publication of Clinical Study Data: https:// trials.boehringer-ingelheim.com/. Prior to providing access, documents will be examined, and, if necessary, redacted and the data will be deidentified, to protect the personal data of study participants and personnel, and to respect the boundaries of the informed consent of the study participants. Clinical Study Reports and Related Clinical Documents can also be requested via the link https://trials.boehringer-ingelheim. com/. All requests will be governed by a Document Sharing Agreement.

Bona fide, qualified scientific and medical researchers may request access to de-identified, analyzable participant clinical study data with corresponding documentation describing the structure and content of the datasets. Upon approval, and governed by a Data Sharing Agreement, data are shared in a secured data-access system for a limited period of 1 year, which may be extended upon request. Researchers should use the https://trials.boehringer-ingelheim.com/ link to request access to study data.

†Address for correspondence: Professor Dragos Vinereanu, MD, PhD, University of Medicine and Pharmacy Carol Davila, University and Emergency Hospital, Splaiul Independentei 169, 050078, Bucharest, Romania. E-mail: vinereanu@gmail.com

This is an open access article distributed under the terms of the Creative Commons Attribution-Non Commercial-No Derivatives License 4.0 (CCBY-NC-ND), where it is permissible to download and share the work provided it is properly cited. The work cannot be changed in any way or used commercially without permission from the journal.

food interactions, and slow onset and offset of action, whereas antiplatelet treatment with clopidogrel or aspirin is ineffective and, therefore, not recommended for stroke prevention in AF by current guidelines.^{8,9} In the past decade, these drawbacks have largely been circumvented by approval of the direct oral anticoagulants (DOACs) for stroke prevention in patients with AF in Europe and around the world.^{8,10} These include the direct thrombin inhibitor, dabigatran, and the factor Xa inhibitors rivaroxaban, apixaban, and edoxaban.¹⁰ In 4 pivotal clinical trials, the efficacy of dabigatran and the factor Xa inhibitors was at least similar to that of VKA for the prevention of stroke or systemic embolism in patients with AF, with reduced risk of intracranial bleeding.^{11–14}

Retrospective analyses of large medical databases support the effectiveness and safety of dabigatran in real-world clinical practice.^{15–18} However, retrospective studies may be compromised by missing or inaccurate data (eg, relating to concomitant use of aspirin)⁸ and, notably, are often limited to individual countries in North America and Western Europe.^{15–18} Although clinical trial and real-world data hint at potential geographical differences in DOAC and VKA treatment outcomes,^{11,19,20} there is a lack of data reported for regions such as Eastern Europe.

Global Registry on Long-Term Oral Antithrombotic Treatment in Patients with Atrial Fibrillation (GLORIA-AF) is a large, prospective, global registry program that, in its third phase, is providing outcome data for dabigatran versus VKA in routine clinical practice.⁸ For the global population of GLORIA-AF, the recently reported final 3-year outcomes in phase 3 demonstrated that dabigatran-treated patients have reduced risk for major bleeding (hazard ratio [HR] 0.61; 95% CI [confidence interval], 0.42–0.88) and all-cause death (HR 0.78; 95%) CI, 0.63–0.97), and similar risk for stroke (HR 0.89; 95%) CI, 0.59–1.34) and myocardial infarction (HR 0.89; 95% CI, 0.53–1.48), compared with VKA-treated patients.²¹ Notably, these outcomes were analyzed with a propensity score (PS)-matched population and multivariable Cox regression.²¹

Herein, we report a descriptive analysis of antithrombotic treatment usage (types and dosages), with the main objective of determining the final 3-year cardiovascular (CV) outcomes for dabigatran and VKA, in the subset of patients from Eastern Europe in phase 3 of GLORIA-AF. Because PS matching has the disadvantage of reducing the number of patients, the descriptive analysis of CV outcomes in the Eastern European subset was not PS matched. To contextualize the Eastern European results, we also descriptively analyze and report the 3-year CV outcomes and baseline characteristics of the global population without PS matching, and compare 3-year persistence with dabigatran in the Eastern European and global populations.

METHODS

Study design and patients

The design of the prospective GLORIA-AF registry program has been reported elsewhere (https://clini caltrials.gov; trial registration numbers NCT01468701, NCT01671007, NCT01937377).

Briefly, data were collected for patients with newly diagnosed AF in routine clinical practice, to evaluate patient characteristics influencing treatment choice, and the effectiveness and safety of antithrombotic therapies.8 Phase 1 was conducted using a cross-sectional approach, before approval of DOACs. Phase 2, which started when dabigatran was approved in participating countries, involved collection of baseline characteristics of patients receiving antithrombotic therapy, with effectiveness and safety outcomes reported for patients receiving dabigatran for 2 years.²² Phase 3 used a cohort study design. Data were gathered for patients prescribed any antithrombotic treatment. To reduce confounding, phase 3 started when the baseline characteristics of patients in phase 2 were similar enough for comprehensive comparative analysis of effectiveness and safety outcomes with dabigatran versus VKA up to 3 years in the global population, based on PS methodology. Findings have recently been published for the PS-matched global population.²¹ No PS matching was performed for comparisons of effectiveness and safety with dabigatran versus VKA in the analyses of the Eastern European and global populations reported here.

In phase 3 of GLORIA-AF, consecutive patients were enrolled between January 2014 and December 2016 in 38 countries in Asia, Europe, North America, and Latin America. Patients were to be recruited from several types of centers, such as general practices, specialist offices, community and university hospitals, outpatient care centers, and anticoagulant clinics. For the current analysis, the Eastern European population was comprised of patients from 9 countries (Bulgaria, Croatia, Czech Republic, Estonia, Latvia, Poland, Romania, Russia, and Slovenia). Detailed patient eligibility criteria have been reported previously.^{8,21} Eligible patients were adults with newly diagnosed nonvalvular AF (<3 months before baseline visit; Latin America <4.5 months because of referral patterns) at risk of stroke (CHA2DS2-VASc score \geq 1). GLORIA-AF was conducted in accordance with the Declaration of Helsinki, other relevant guidelines, and the study protocol. All patients provided written informed consent.

Objectives and outcomes

The main objective of this descriptive analysis was to determine the 3-year CV outcomes for dabigatran and VKA in the subset of patients from Eastern Europe, with comparisons to the global population. The outcomes were: stroke (hemorrhagic, ischemic, or uncertain classification); major bleeding, defined using International Society on Thrombosis and Haemostasis criteria²³; all-cause death; and a CV composite of stroke, systemic embolism, myocardial infarction, life-threatening bleeding, and vascular death. To speculate on reasons for potential differences in CV outcomes between the Eastern European and global populations, we compared the baseline characteristics of patients in the 2 populations.

Given that patients were eligible for Phase 3 of GLORIA-AF regardless of antithrombotic therapy, we also reported treatment patterns for patients receiving any or no antithrombotic therapy.

Treatment persistence with dabigatran was also reported for the Eastern European and global populations, defined as remaining on therapy without interruption for longer than 30 days.

Statistical analysis

Baseline characteristics, demographics, treatment patterns (including dosage), and incidence rates of all key CV outcomes were summarized descriptively for all eligible patients in the Eastern European and global populations. Regarding the descriptive analysis of CV outcomes, incidence rates per 100 patient-years, with 95% CIs, were calculated in the dabigatran and VKA treatment groups, based on data from all eligible patients (excluding those not treated with prescribed antithrombotic therapy) in the Eastern European and global populations. The case report form was set up in such a way that investigators had to enter all outcome event information and changes in medication since the last study visit; therefore, complete information on outcome events was ensured even if intermittent visits were missed, using all available information from the completed study visits.

Treatment persistence with dabigatran at 6, 12, 24, and 36 months was estimated using the Kaplan–Meier method, based on data from all eligible patients (excluding those not treated with prescribed antithrombotic therapy) in the Eastern European and global populations.

Multivariable Cox regression analysis was prespecified only on the PS-matched global population²¹; no statistical testing was performed on the treatment outcomes for the current analyses. All analyses were performed using SAS[®] software version 9.4 or later (SAS Institute, Inc., Cary, NC).

RESULTS

Patient disposition and antithrombotic treatments

In Eastern Europe, 1341 patients were eligible for analysis, comprising 6% of 21,300 eligible patients enrolled between January 2014 and December 2016 in the global population. Eastern European patients were from Bulgaria (22%; N = 296), Romania (22%; N = 295), Poland (18%; N = 239), Russia (15%; N = 195), Croatia (10%; N = 139), Czech Republic (6%; N = 86), Slovenia (3%; N = 37), Latvia (2%; N = 29), and Estonia (2%; N = 25). In addition to Eastern Europe, the global population included patients from Western Europe (43%; N = 9158), North America (24%; N = 5120), Asia (19%; N = 4053), and Latin America (8%; N = 1628).

Of the 1341 eligible patients in Eastern Europe, 80.8% completed the planned 3-year observation period (range: 74.2% [VKA] to 85.7% [dabigatran]), compared with 80.5% in the global population (range: 77.6% [antiplatelets or no antithrombotic treatment] to 83.8% [dabigatran]) (see **Figure**, **Supplemental Digital Content 1**, http://links.lww.com/AJT/A142).

The distribution of prescribed antithrombotic treatments differed in Eastern Europe versus the global population (Figure 1). Dabigatran and VKA were the most prescribed antithrombotic treatments in the Eastern European population, more than in the global population (dabigatran 37.5% vs. 18.0%, VKA 35.2% vs. 22.7%). Factor Xa inhibitors were prescribed less often in the Eastern European population than in the global population (rivaroxaban 10.9% vs. 18.9%, apixaban 3.2% vs. 21.2%, edoxaban 0% vs. 1.6%). Antiplatelet agents (without concomitant dabigatran, VKA, or factor Xa inhibitors) were prescribed in 10.7% of the Eastern European population, similar to the global population (11.2%). Only 2.5% of the Eastern European population was prescribed no antithrombotic treatment, compared with 6.5% of the global population. As shown in Supplemental Digital Content 1, (see Table, http://links.lww.com/AJT/A142), there were large differences in prescriptions of antithrombotic treatments between countries; for instance, 54.0% of patients in Poland (n = 129/239) were prescribed dabigatran, compared with only 5.8% (n = 5/ 86) in the Czech Republic.

In Eastern Europe, 70% of patients were prescribed the standard dose (150 mg twice daily [BID]) and 29% were prescribed the lower dose of dabigatran (110/75 mg BID), compared with 52% and 46% in the global population, respectively (Figure 2). By contrast, the proportions of patients prescribed standard or lower



FIGURE 1. Baseline antithrombotic treatment prescriptions in the Eastern European and global populations. ^aThe denominator for Eastern Europe is the total number of eligible patients in the Eastern European population (N = 1341). The denominator for the global population is the total number of eligible patients in the global population (N = 21,300). Tx, treatment; VKA, vitamin K antagonist.

doses of the factor Xa inhibitors were comparable in Eastern Europe versus the global population.

Patient characteristics

As shown in Table 1, Eastern European patients' demographics (mean age, 67.8 years [SD, 10.1]; body mass index [BMI], 29.1 [SD, 4.7]; 50.0% female; 99.0% White) and disease characteristics were generally well balanced across the treatment groups. A notable exception is that 78.0% of patients in the 'other' group experienced paroxysmal AF, relative to 40.0%–54.5% across the DOAC and VKA groups. Patients in the 'other' group received high bleeding risk medications such as antiplatelet agents (n = 144) or no antithrombotic treatment (n = 33). CHA2DS2-VASc scores were comparable, ranging from 3.0 to 3.4, across the treatment groups.

For comparison with the Eastern European population, patient characteristics for all eligible patients in the global population are also shown in Table 1. At baseline, chronic concomitant use of medications associated with a high risk of bleeding, including antiplatelets, was substantially lower in Eastern Europe (16.4%) than globally (28.3%). Consistent with this observation, mean HAS-BLED scores were lower in Eastern Europe (1.1 [SD, 0.8]) compared with the global population (1.4 [SD, 0.9]), and history of bleeding events was also less common in Eastern Europe (3.1%) than globally (5.3%). Although CHA2DS2VASc scores did not differ, the proportion of patients with history of stroke was lower in Eastern Europe (7.3%) than globally (10.6%). Conversely, in Eastern Europe, congestive heart failure was more common (37.7%) than in the global population (21.7%). This was also the case for hypertension (Eastern Europe, controlled 72.9%, uncontrolled 13.7%; globally, controlled 62.9%, uncontrolled 10.2%).

At baseline, chronic concomitant use of medications associated with a high risk of bleeding was lower in the dabigatran group than VKA group in Eastern Europe (3.8% vs. 9.3%) and globally (14.8% vs. 20.6%). HAS-BLED and CHA2DS2-VASc scores, and history of bleeding and stroke were generally comparable in the dabigatran and VKA groups in Eastern Europe and globally (Table 2).

Effectiveness and safety of dabigatran versus VKA

In Eastern Europe, the incidence rates for major bleeding (0.26 vs. 0.90), all-cause death (2.04 vs. 3.50), and the CV composite outcome (1.37 vs. 1.92) were numerically lower with dabigatran versus VKA, and the incidence rates for stroke (0.51 vs. 0.50) were comparable for the 2 treatment groups (Figure 3). Although these incidence rates were consistent with the beneficial impact of dabigatran versus VKA in the global population, the rates of all 4 outcomes were lower in Eastern Europe versus the global population for both



FIGURE 2. Baseline DOAC prescription dosage in the Eastern European and global populations. ^aThe denominators are all eligible patients (ie, the ns shown per treatment group). No patients in Eastern Europe were prescribed or treated with edoxaban. ^bDabigatran standard dose (150 mg BID), lower dose (110/75 mg BID); rivaroxaban standard dose (20 mg QD), lower dose (15/10 mg QD); apixaban standard dose (5 mg BID), lower dose (2.5 mg BID); edoxaban standard dose (60 mg QD), lower dose (30 mg QD); "other" doses were any dose that did not fit under the definitions of standard or lower doses per treatment. BID, twice daily; DOAC, direct oral anticoagulants; QD, once daily.

treatments (Figure 3). However, the numbers of patients (overall and with CV events) were substantially lower in Eastern Europe versus the global population.

Persistence with dabigatran treatment

Based on the Kaplan–Meier method, 36-month probabilities of dabigatran persistence (ie, without treatment interruptions longer than 30 days) were 76% and 64% in the Eastern European and global populations, respectively (Figure 4). Estimates of dabigatran treatment persistence were also higher in Eastern Europe versus the global population at all earlier time points: 89% versus 81% at 6 months, 84% versus 74% at 12 months, and 79% versus 68% at 24 months.

DISCUSSION

In this descriptive analysis of the GLORIA-AF registry, 3 years of treatment with dabigatran was associated with numerically reduced risks of major bleeding, all-cause death, and a composite CV outcome, and comparable risk of stroke, versus VKA in Eastern European patients with newly diagnosed AF. Definitive conclusions about the effectiveness and safety of dabigatran and VKA in this Eastern European population are prevented by the lack of statistical testing, small numbers of patients with CV events (eg, in the dabigatran and VKA groups combined, 11 patients vs. 158 patients in the global population experienced stroke, respectively), and no PS matching. Undetected factors could bias the findings, and causality cannot be attributed to the antithrombotic treatments. However, the findings in Eastern Europe are consistent with the beneficial impact of dabigatran versus VKA in the substantially larger global population, previously reported with PS matching and statistical testing²¹ and reported herein without PS matching.

Regardless of treatment (dabigatran or VKA), the incidence rates of all 4 outcomes at 3 years were numerically lower in Eastern Europe than globally. Three-year dabigatran treatment persistence was higher (76% and 64%, respectively) and use of chronic concomitant high bleeding risk therapies at baseline was lower in Eastern Europe than globally, possibly contributing to the good outcomes in the Eastern European population. There could also be an element of chance, owing to the small number of patients experiencing these CV events in the Eastern European versus global population.

In the Eastern European population, which was enrolled up to December 2016, a higher proportion (78.0%) of patients in the "other" group (receiving high bleeding risk therapies without concomitant DOACs or VKA, or no antithrombotic treatment) experienced paroxysmal AF than across the DOAC and VKA groups

Table 1.	Baseline	characteristics	of eligible	patients	in the Eastern	European	and global	populations
----------	----------	-----------------	-------------	----------	----------------	----------	------------	-------------

	Eastern Europe					Global
	Dabigatran (N = 503)	VKA (N = 472)	Factor Xa inhibitor* (N = 189)	Other† (N = 177)	Total (N = 1341)	Total (N = 21,300)
Age (yr), mean (SD)	67.0 (10.1)	69.2 (9.2)	68.2 (10.4)	66.1 (11.6)	67.8 (10.1)	70.5 (10.6)
65–74, n (%)	193 (38.4)	184 (39.0)	66 (34.9)	58 (32.8)	501 (37.4)	7712 (36.2)
≥75, n (%)	128 (25.4)	146 (30.9)	57 (30.2)	45 (25.4)	376 (28.0)	8166 (38.3)
Female, n (%)	250 (49.7)	234 (49.6)	95 (50.3)	92 (52.0)	671 (50.0)	9568 (44.9)
Race, n (%)‡						
Asian	0	0	0	0	0	4132 (19.4)
Black/African American	2 (0.4)	4 (0.8)	0	1 (0.6)	7 (0.5)	398 (1.9)
White	498 (99.0)	466 (98.7)	188 (99.5)	176 (99.4)	1328 (99.0)	14,823 (69.6)
BMI (kg/m²), mean (SD)	29.1 (4.8)	29.2 (4.7)	28.9 (4.8)	28.6 (4.6)	29.1 (4.7)	28.6 (6.4)
CrCl (mL/min), mean (SD)	86.6 (32.0)	80.1 (30.6)	81.3 (32.9)	85.9 (37.9)	83.5 (32.6)	81.4 (60.5)
HAS-BLED score, mean (SD)	0.9 (0.7)	1.0 (0.8)	1.0 (0.8)	1.7 (0.9)	1.1 (0.8)	1.4 (0.9)
CHA ₂ DS ₂ -VASc score, mean (SD)	3.1 (1.4)	3.3 (1.4)	3.4 (1.6)	3.0 (1.4)	3.2 (1.4)	3.2 (1.5)
Type of AF, n (%)						
Paroxysmal	225 (44.7)	189 (40.0)	103 (54.5)	138 (78.0)	655 (48.8)	12,001 (56.3)
Persistent	216 (42.9)	220 (46.6)	68 (36.0)	32 (18.1)	536 (40.0)	7277 (34.2)
Permanent	62 (12.3)	63 (13.3)	18 (9.5)	7 (4.0)	150 (11.2)	2022 (9.5)
Chronic concomitant medications, n (%)						
Antiplatelet	13 (2.6)	32 (6.8)	9 (4.8)	22 (12.4)	76 (5.7)	5558 (26.1)
High bleeding risk medications§	19 (3.8)	44 (9.3)	17 (9.0)	140 (79.1)	220 (16.4)	6034 (28.3)
Medical history, n (%)						
Controlled hypertension	363 (72.2)	344 (72.9)	146 (77.2)	124 (70.1)	977 (72.9)	13,404 (62.9)
Uncontrolled hypertension	76 (15.1)	63 (13.3)	18 (9.5)	27 (15.3)	184 (13.7)	2162 (10.2)
Stroke	37 (7.4)	30 (6.4)	17 (9.0)	14 (7.9)	98 (7.3)	2260 (10.6)
Transient ischemic attack	17 (3.4)	12 (2.5)	8 (4.2)	3 (1.7)	40 (3.0)	952 (4.5)
Deep vein thrombosis	6 (1.2)	2 (0.4)	0	0	8 (0.6)	248 (1.2)
Coronary artery disease	76 (15.1)	91 (19.3)	45 (23.8)	45 (25.4)	257 (19.2)	4000 (18.8)
Angina pectoris	65 (12.9)	79 (16.7)	28 (14.8)	33 (18.6)	205 (15.3)	1964 (9.2)
Myocardial infarction	28 (5.6)	48 (10.2)	21 (11.1)	14 (7.9)	111 (8.3)	2062 (9.7)
Congestive heart failure	192 (38.2)	198 (41.9)	66 (34.9)	50 (28.2)	506 (37.7)	4632 (21.7)
Rheumatic heart disease	1 (0.2)	4 (0.8)	1 (0.5)	0	6 (0.4)	92 (0.4)
Peripheral artery disease	5 (1.0)	14 (3.0)	6 (3.2)	7 (4.0)	32 (2.4)	623 (2.9)
Bleeding	12 (2.4)	9 (1.9)	15 (7.9)	6 (3.4)	42 (3.1)	1130 (5.3)
Pulmonary embolism	2 (0.4)	1 (0.2)	2 (1.1)	0	5 (0.4)	125 (0.6)
Non-CNS arterial embolism	3 (0.6)	1 (0.2)	0	0	4 (0.3)	77 (0.4)
Complex aortic plaque	8 (1.6)	8 (1.7)	6 (3.2)	4 (2.3)	26 (1.9)	205 (1.0)
Diabetes	102 (20.3)	98 (20.8)	55 (29.1)	39 (22.0)	294 (21.9)	4960 (23.3)
Hyperlipidaemia	198 (39.4)	201 (42.6)	74 (39.2)	82 (46.3)	555 (41.4)	8340 (39.2)
Hepatic disease	9 (1.8)	10 (2.1)	2 (1.1)	4 (2.3)	25 (1.9)	310 (1.5)
Abnormal kidney function	1 (0.2)	7 (1.5)	1 (0.5)	2 (1.1)	11 (0.8)	392 (1.8)

*Rivaroxaban or apixaban.

†Aspirin, antiplatelets other than aspirin, or no antithrombotic treatment.

‡In the Eastern European and global populations, patients also identified as Native American/Alaskan/Hawaiian/Other Pacific Islander (0.5%, 0.6%), other (0.1%, 3.0%), or a racial designation was not available (0.3%, 5.5%), respectively.

\$Risk based on the HAS-BLED score (antiplatelet agent, Cox-2 inhibitor, or other nonsteroidal anti-inflammatory drug).

AF, atrial fibrillation; BMI, body mass index; CNS, central nervous system; CrCL, creatinine clearance SD, standard deviation; VKA, vitamin K antagonist.

www.americantherapeutics.com

	Eastern Eu	rope	Global		
	Dabigatran (N = 503)	VKA (N = 472)	Dabigatran (N = 3839)	VKA (N = 4836)	
Age (yr), mean (SD)	67.0 (10.1)	69.2 (9.2)	70.1 (10.2)	71.2 (10.3)	
Female, n (%)	250 (49.7)	234 (49.6)	1718 (44.8)	2152 (44.5)	
Race, n (%)†					
Asian	0	0	841 (21.9)	760 (15.7)	
Black/African American	2 (0.4)	4 (0.8)	47 (1.2)	85 (1.7)	
White	498 (99.0)	466 (98.7)	2484 (64.7)	3611 (74.7)	
CrCl (mL/min), mean (SD)	86.6 (32.0)	80.1 (30.6)	83.5 (117.4)	76.8 (35.4)	
HAS-BLED score, mean (SD)	0.9 (0.7)	1.0 (0.8)	1.2 (0.8)	1.3 (0.9)	
CHA ₂ DS ₂ -VASc score, mean (SD)	3.1 (1.4)	3.3 (1.4)	3.1 (1.4)	3.3 (1.5)	
Type of AF, n (%)					
Paroxysmal	225 (44.7)	189 (40.0)	2082 (54.2)	2174 (45.0)	
Persistent	216 (42.9)	220 (46.6)	1309 (34.1)	1977 (40.9)	
Permanent	62 (12.3)	63 (13.3)	448 (11.7)	685 (14.2)	
Chronic concomitant medications, n (%)					
Antiplatelet	13 (2.6)	32 (6.8)	508 (13.2)	913 (18.9)	
High bleeding risk medications‡	19 (3.8)	44 (9.3)	569 (14.8)	998 (20.6)	
Medical history, n (%)					
Controlled hypertension	363 (72.2)	344 (72.9)	2416 (62.9)	3104 (64.2)	
Uncontrolled hypertension	76 (15.1)	63 (13.3)	421 (11.0)	467 (9.7)	
Congestive heart failure	192 (38.2)	198 (41.9)	749 (19.5)	1374 (28.4)	
Diabetes	102 (20.3)	98 (20.8)	828 (21.6)	1233 (25.5)	
Stroke	37 (7.4)	30 (6.4)	441 (11.5)	462 (9.6)	
Coronary artery disease	76 (15.1)	91 (19.3)	511 (13.3)	916 (18.9)	
Bleeding	12 (2.4)	9 (1.9)	138 (3.6)	251 (5.2)	

Table 2. Selected* baseline characteristics of eligible patients in the dabigatran and VKA groups in the Eastern European and global populations.

*Selected from the characteristics in Table 1.

†In the Eastern European population (dabigatran vs. VKA), patients also identified as Native American/Alaskan (0.2% vs. 0%) or a racial designation was not available (0.4% vs. 0.4%), respectively. In the global population (dabigatran vs. VKA), patients also identified as Native American/Alaskan (0.7% vs. 0.7%), other (4.7% vs. 3.0%), or a racial designation was not available (6.7% vs. 4.2%), respectively.
‡Risk based on the HAS-BLED score (antiplatelet agent, Cox-2 inhibitor, or other nonsteroidal anti-inflammatory drug).

AF, atrial fibrillation; CNS, central nervous system; CrCL, creatinine clearance; SD, standard deviation; VKA, vitamin K antagonist.

(40.0%–54.5%), despite comparable CHA2DS2-VASc scores (3.0–3.4). Although the 2020 ESC guidelines state that AF type "should not condition the indication to thromboprophylaxis,"⁹ choice of treatment may have been influenced by the short-term nature of paroxysmal AF.

Consistent with the higher three-year persistence with dabigatran treatment in Eastern Europe than globally, previous GLORIA-AF papers have suggested regional differences in persistence with DOACs, including dabigatran, up to 2 years of treatment.^{19,20,24,25} Factors that predicted discontinuation of DOACs during 1-year and 2-year follow-ups of GLORIA-AF included geographical region (Asia and North America vs. Europe) and, perhaps counterintuitively, private versus federal insurance.^{24,25}

In the first 2 years of therapy, several investigations have reported high persistence rates for DOACs, including dabigatran versus VKA.^{24,26–29} The current study extends these findings, demonstrating high persistence with 3 years of dabigatran treatment. Reasons for high levels of compliance with DOACs likely include ease of use, including no requirement to monitor DOAC levels except for special cases such as in patients with particularly high or low body weight.^{30–}



FIGURE 3. Incidence rates of important clinical outcomes during 3 years of treatment with dabigatran versus VKA in Eastern Europe versus the global population. Each n denotes the number of patients with events. Results are shown for all eligible patients treated with dabigatran and VKA in Eastern Europe (N = 498 and N = 466, respectively) and globally (N = 3807 and N = 4788, respectively), that is, excluding a few patients with a prescription but not treated. ^aStroke, systemic embolism, myocardial infarction, life-threatening bleed, vascular death CI, confidence interval; VKA, vitamin K antagonist.

³³ The advantage of lack of necessity to monitor anticoagulation with DOACs may improve patients' access to treatment (eg, for those in remote rural areas). DOAC plasma levels also decrease quickly,³³ allowing more straightforward surgical interventions.

In the current study, the beneficial impact of dabigatran versus VKA on the incidence of CV outcomes was observed with different proportions of patients receiving the standard dabigatran dose (70%, 150 mg BID) and lower dose of dabigatran (29%, 110/75 mg BID) in Eastern Europe than in the global population (52% and 46%, respectively). This is notable, given that some differences in CV outcomes between and within studies may be explained by dabigatran dosing. In the RE-LY phase 3 trial (N = 18,113), dabigatran 150 mg BID was more beneficial than warfarin for stroke prevention (relative risk [RR] 0.64; 95% CI, 0.51–0.81), whereas 110 mg BID was comparable to warfarin (RR, 0.92; 95% CI, 0.74-1.13). Conversely, dabigatran 110 mg BID was more beneficial than warfarin for major bleeding (RR 0.80; 95% CI, 0.69–0.93), whereas 150 mg BID was comparable to warfarin (RR 0.93; 95% CI, 0.81-1.07).¹¹ In 2 large US claims database studies, dabigatran was associated with lower incidence of stroke and fewer intracranial bleeding events than with warfarin,^{15,17} particularly with the 150-mg dose of dabigatran, whereas the 75-mg dose was associated with no difference in stroke prevention and reduced intracranial hemorrhage versus warfarin.¹⁷

Although our descriptive analysis suggests that antithrombotic treatment patterns differ in Eastern Europe relative to the global population, these patterns will have changed since the last patients were enrolled into GLORIA-AF phase 3 in December 2016. A survey in the Balkan region (N = 2663) demonstrated greater use of VKA than in our study (60.9% vs. 35.2%), and increasing use of dabigatran and factor Xa inhibitors during a 14-week period between December 2014 and February 2015.^{34,35} Although changes in clinical practice will have been influenced by research published in the intervening years, OAC use in the Balkan region was largely guided by factors other than evidence-based medicine, such as drug availability and reimbursement policy.³⁵ Clinical practices have benefitted in recent years from greater awareness and education, including greater communication between cardiology subspecialities, neurology, and general practice. Even though the situation has improved with the introduction of DOACs in the last decade, there is still an issue with drug-drug interactions and patients taking too many medicines at full dosage, such as nonsteroidal anti-inflammatory drugs, antiplatelets, and OACs, in combination regimens. The issue of drug-drug interactions is less common with DOACs than VKA.³⁰ However, it is still important to optimize DOACs and other treatment regimens for efficacy and safety (eg, for kidney function, leading to potential adaptation of dosage), which, in



FIGURE 4. Probability of persistence with dabigatran during 3 years of treatment in Eastern Europe versus the global population. Results are shown for all eligible patients treated with dabigatran and VKA in Eastern Europe (N = 498 and N = 466, respectively) and globally (N = 3807 and N = 4788, respectively), that is, excluding a few patients with a prescription but not treated. CI, confidence interval; VKA, vitamin K antagonist.

our experience, may be achieved via multifaceted and multilevel educational intervention.³⁶

Strengths and limitations

Our study has several important strengths and limitations. Although dabigatran was associated with treatment outcomes that were comparable or superior to those with VKA, the study was not designed to prove causality and, notably, no PS matching or statistical tests were applied. However, our findings are compatible with those previously reported for the PS-matched global population, analyzed by multivariable Cox regression.²¹ PS matching has the disadvantage of reducing the number of patients in the analyzed population and, notably, our Eastern European population had a small number of patients with CV events (eg, in the dabigatran and VKA groups combined, 11 patients vs. 158 patients in the global population experienced stroke). Similarly, for the Eastern European population, the small number of patients treated with factor Xa inhibitors (N = 188) precluded comparisons of their effectiveness and safety with dabigatran (N = 498) and VKA (N = 466). For the global population, post hoc analyses suggest that patients treated with dabigatran may benefit from a reduced (by 41%) or similar risk of major bleeding versus rivaroxaban and apixaban, respectively, and that all 3 treatments had similar risk of stroke, myocardial infarction, and death.³⁷ Another potential drawback is that treatment persistence may be enhanced by frequent clinical follow-up. Conversely, regular follow-up with physicians, and the associated procedures to gather and review data (eg, 10% on-site monitoring) have resulted in data that are of particularly high quality for a real-world study. The 3-year follow-up period is also notable, given that there is a high risk of discontinuing antithrombotic therapy before 1 year,^{24,25,38} and limited information for the effectiveness, persistence, and safety beyond 2 years of treatment with DOACs in real-world settings.

Summary

In summary, in this 3-year descriptive analysis of the prospective GLORIA-AF registry, dabigatran was associated with numerically reduced risk of major bleeding, all-cause death, and a composite CV outcome, whereas the risk of stroke was comparable, relative to VKA, in the Eastern European and global populations. The Eastern European population, compared with the global population, was associated with substantially less use of chronic concomitant high bleeding risk therapies at baseline and substantially higher persistence with dabigatran treatment. The incidence of CV outcomes with dabigatran, compared with VKA, in the current descriptive analysis are compatible with those in other comparative analyses including the pivotal RE-LY trial, retrospective studies, and analyses performed in regions other than Eastern Europe.^{11,15,17}

Our 3-year descriptive analysis of CV outcomes from GLORIA-AF phase 3 is limited by the lack of statistical testing or PS-matching and the small numbers of patients with CV events (eg, in the dabigatran and VKA groups combined, 11 patients experienced stroke), preventing definitive conclusions about the effectiveness and safety of antithrombotic treatments in the Eastern European population. However, comparison with the CV outcomes in the substantially larger global population (previously reported with PS matching and statistical testing²¹ and reported herein without PS matching) suggests that dabigatran versus VKA has a favorable benefit-risk profile in routine clinical practice in Eastern European patients with newly diagnosed AF.

ACKNOWLEDGMENTS

Editorial support was provided by Fortis Pharma Consulting, with financial support by Boehringer Ingelheim.

REFERENCES

- Haim M, Hoshen M, Reges O, et al. Prospective national study of the prevalence, incidence, management and outcome of a large contemporary cohort of patients with incident non-valvular atrial fibrillation. *J Am Heart Assoc*. 2015;4:e001486.
- Lloyd-Jones DM, Wang TJ, Leip EP, et al. Lifetime risk for development of atrial fibrillation: the Framingham Heart Study. *Circulation*. 2004;110:1042–1046.
- 3. Rahman F, Kwan GF, Benjamin EJ. Global epidemiology of atrial fibrillation. *Nat Rev Cardiol*. 2014;11:639–654.
- 4. Wolf PA, Abbott RD, Kannel WB. Atrial fibrillation as an independent risk factor for stroke: the Framingham Study. *Stroke*. 1991;22:983–988.
- Lee E, Choi EK, Han KD, et al. Mortality and causes of death in patients with atrial fibrillation: a nationwide population-based study. Novo G, ed. *PLoS One.* 2018; 13:e0209687.
- Jani BD, Nicholl BI, McQueenie R, et al. Multimorbidity and co-morbidity in atrial fibrillation and effects on survival: findings from UK Biobank cohort. *EP Eur.* 2018;20: f329–f336.
- Hart RG, Pearce LA, Aguilar MI. Meta-analysis: antithrombotic therapy to prevent stroke in patients who have nonvalvular atrial fibrillation. *Ann Intern Med.* 2007;146:857.
- Huisman MV, Lip GYH, Diener HC, et al. Design and rationale of Global Registry on Long-Term Oral Antithrombotic Treatment in Patients with Atrial Fibrillation: a global registry program on long-term oral antithrombotic treatment in patients with atrial fibrillation. *Am Heart J.* 2014;167:329–334.
- 9. Hindricks G, Potpara T, Dagres N, et al. 2020 ESC Guidelines for the diagnosis and management of atrial fibrillation developed in collaboration with the European

Association for Cardio-Thoracic Surgery (EACTS). Eur Heart J. 2021;42:373–498.

- 10. Kozieł M, Teutsch C, Bayer V, et al. Changes in anticoagulant prescription patterns over time for patients with atrial fibrillation around the world. *J Arrhythmia*. 2021;37: 990–1006.
- 11. Connolly SJ, Ezekowitz MD, Yusuf S, et al. Dabigatran versus warfarin in patients with atrial fibrillation. *N Engl J Med*. 2009;361:1139–1151.
- Patel MR, Mahaffey KW, Garg J, et al. Rivaroxaban versus warfarin in nonvalvular atrial fibrillation. N Engl J Med. 2011;365:883–891.
- Granger CB, Alexander JH, McMurray JJV, et al. Apixaban versus warfarin in patients with atrial fibrillation. N Engl J Med. 2011;365:981–992.
- Giugliano RP, Ruff CT, Braunwald E, et al. Edoxaban versus warfarin in patients with atrial fibrillation. *N Engl J Med*. 2013;369:2093–2104.
- 15. Villines TC, Schnee J, Fraeman K, et al. A comparison of the safety and effectiveness of dabigatran and warfarin in non-valvular atrial fibrillation patients in a large healthcare system. *Thromb Haemost.* 2015;114: 1290–1298.
- Larsen TB, Gorst-Rasmussen A, Rasmussen LH, et al. Bleeding events among new starters and switchers to dabigatran compared with warfarin in atrial fibrillation. *Am J Med.* 2014;127:650–656.e5.
- 17. Graham DJ, Reichman ME, Wernecke M, et al. Cardiovascular, bleeding, and mortality risks in elderly medicare patients treated with dabigatran or warfarin for nonvalvular atrial fibrillation. *Circulation*. 2015;131:157–164.
- 18. Graham DJ, Baro E, Zhang R, et al. Comparative stroke, bleeding, and mortality risks in older medicare patients treated with oral anticoagulants for nonvalvular atrial fibrillation. *Am J Med.* 2019;132:596–604.e11.
- 19. Dubner S, Saraiva JFK, Fragoso JCN, et al. Effectiveness and safety of dabigatran in Latin American patients with atrial fibrillation: two years follow up results from GLORIA-AF registry. *IJC Heart Vasc.* 2020;31:100666.
- 20. Azar RR, Ragy HI, Kozan O, et al. Antithrombotic treatment pattern in newly diagnosed atrial fibrillation patients and 2-year follow-up results for dabigatrantreated patients in the Africa/Middle-East Region: phase II results from the GLORIA-AF registry program. *IJC Heart Vasc.* 2021;34:100763.
- Huisman MV, Teutsch C, Lu S, et al. Dabigatran versus vitamin K antagonists for atrial fibrillation in clinical practice: final outcomes from Phase III of the GLORIA-AF registry. *Clin Res Cardiol.* 2022;111:548–559.
- 22. Huisman MV, Rothman KJ, Paquette M, et al. Two-year follow-up of patients treated with dabigatran for stroke prevention in atrial fibrillation: Global Registry on Long-Term Antithrombotic Treatment in Patients with Atrial Fibrillation (GLORIA-AF) registry. *Am Heart J.* 2018;198: 55–63.
- 23. Schulman S, Angerås U, Bergqvist D, et al. Definition of major bleeding in clinical investigations of antihemostatic medicinal products in surgical patients. *J Thromb Haemost*. 2010;8:202–204.

www.americantherapeutics.com

- 24. Kozieł M, Mazurek M, Teutsch C, et al. Persistence with anticoagulation for atrial fibrillation: report from the GLORIA-AF phase III 1-year follow-up. *J Clin Med.* 2020;9:1969.
- 25. Paquette M, França LR, Teutsch C, et al. Dabigatran persistence and outcomes following discontinuation in atrial fibrillation patients from the GLORIA-AF registry. *Am J Cardiol.* 2020;125:383–391.
- 26. Martinez C, Katholing A, Wallenhorst C, et al. Therapy persistence in newly diagnosed non-valvular atrial fibrillation treated with warfarin or NOAC: a cohort study. *Thromb Haemost.* 2016;115:31–39.
- 27. Zalesak M, Siu K, Francis K, et al. Higher persistence in newly diagnosed nonvalvular atrial fibrillation patients treated with dabigatran versus warfarin. *Circ Cardiovasc Qual Outcomes*. 2013;6:567–574.
- 28. Laliberté F, Cloutier M, Nelson WW, et al. Real-world comparative effectiveness and safety of rivaroxaban and warfarin in nonvalvular atrial fibrillation patients. *Curr Med Res Opin*. 2014;30:1317–1325.
- 29. Nelson WW, Song X, Coleman CI, et al. Medication persistence and discontinuation of rivaroxaban versus warfarin among patients with non-valvular atrial fibrillation. *Curr Med Res Opin*. 2014;30:2461–2469.
- 30. Dunois C. Laboratory monitoring of direct oral anticoagulants (DOACs). *Biomedicines*. 2021;9:445.
- Meyre P, Conen D, Osswald S, et al. Atrial fibrillation for internists: current practice. *Swiss Med Wkly*. 2020;150: w20196.

- 32. Choi EK, Lee YS, Chern AKC, et al. Real-world evaluation of perception, convenience and anticoagulant treatment satisfaction of patients with atrial fibrillation switched from long-term vitamin K antagonist treatment to dabigatran. *Open Heart.* 2020;7:e001343.
- 33. Pollack CV. Coagulation assessment with the new generation of oral anticoagulants. *Emerg Med J.* 2016;33: 423–430.
- 34. Potpara TS, Dan GA, Trendafilova E, et al. Stroke prevention in atrial fibrillation and 'real world' adherence to guidelines in the Balkan Region: the BALKAN-AF Survey. *Sci Rep.* 2016;6:20432.
- 35. Potpara TS, Trendafilova E, Dan GA, et al. The patterns of non-vitamin K antagonist oral anticoagulants (NOACs) use in patients with atrial fibrillation in seven Balkan countries: a report from the BALKAN-AF survey. *Adv Ther*. 2017;34:2043–2057.
- Vinereanu D, Lopes RD, Bahit MC, et al. A multifaceted intervention to improve treatment with oral anticoagulants in atrial fibrillation (IMPACT-AF): an international, cluster-randomised trial. *Lancet*. 2017; 390:1737–1746.
- 37. Lip GYH, Kotalczyk A, Teutsch C, et al. Comparative effectiveness and safety of non-vitamin K antagonists for atrial fibrillation in clinical practice: GLORIA-AF Registry. *Clin Res Cardiol*. 2022;111:560–573.
- 38. Paquette M, Riou França L, Teutsch C, et al. Persistence with dabigatran therapy at 2 years in patients with atrial fibrillation. *J Am Coll Cardiol.* 2017;70:1573–1583.